Author's response to reviews

Title: Comprehensive analysis of NuMA variation in breast cancer

Authors:

Outi Kilpivaara (outi.kilpivaara@helsinki.fi)
Matias Rantanen (matias.rantanen@helsinki.fi)
Anitta Tamminen (anitta.tamminen@hus.fi)
Kristiina Aittomaki (kristiina.aittomaki@hus.fi)
Carl Blomqvist (carl.blomqvist@hus.fi)
Heli Nevanlinna (heli.nevanlinna@hus.fi)

Version: 5 Date: 19 February 2008

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

Thank you for accepting our manuscript (Comprehensive analysis of NuMA variation in breast cancer MS: 3220913101567250) to be published in BMC Cancer. You requested some minor formatting changes and I have done my best to fulfill those.

The changes made to the manuscript are as follows:

Title : the unnecessary capitalizations have been removed

Affiliations : each affiliation is now written in full (department, institute, city, and country).

Introduction : intro has been renamed 'Background'.

Materials and Methods : M&M has been renamed 'Methods'

I hope our manuscript is now ready for publication.

Best regards, Outi Kilpivaara